Author's response to reviews

Title: Perceptions of health care providers and patients on quality of care in maternal and neonatal health in fourteen Bangladesh government healthcare facilities: a mixed-method study

Authors:

Farzana Islam (farzana.islam@oru.se)
Aminur Rahman (aminur@ciprb.org)
Abdul Halim (halim.ogsb@gmail.com)
Charli Eriksson (Charli.Eriksson@oru.se)
Fazlur Rahman (fazlur@ciprb.org)
Koustuv Dalal (koustuv.dalal@oru.se)

Version: 5 Date: 18 May 2015

Author's response to reviews: see over
To
The Editor
BMC Health Services Research

Dear Sir
I am submitting the revised manuscript titled “Perceptions of health care providers and patients on quality of care in maternal and neonatal health in fourteen Bangladesh government healthcare facilities: a mixed-method study” (MS ID: 1079450313154507) on behalf of all the co-authors to publish in your journal.

We have incorporated most of the suggestions made by the reviewers. It is worth mentioning that to improve the quality of English language the manuscript was copyedited by Edanz. Please note that the title of the revised manuscript has been slightly changed.

The paper is original and has not been submitted elsewhere for publication. All authors read the revised version and agreed to submit it for publication.

It will be highly appreciated if you could consider publishing the manuscript in your journal.

Corresponding author
Dr. Farzana Islam, MD, MPH
Doctoral Fellow, School of Health and Medical Sciences, Örebro University, Örebro, Sweden
&
Research Associate, Centre for Injury Prevention and Research, Bangladesh (CIPRB)
House: B-162, Road- 23, Mohakhali DOHS, Dhaka- 1206, Bangladesh.
E-mail: farzana.islam@oru.se
Tel  : +8802-8814988
Fax : +8802-8814964
Telephone: +880-2-8814988; Fax: +880-2-8814964